Author's response to reviews

Title: Developing Optimal Search Strategies for Detecting Clinically Sound Prognostic Studies in MEDLINE: An Analytic Survey

Authors:

Nancy L Wilczynski (wilczyn@mcmaster.ca)
Prof R. BRIAN Haynes (bhaynes@mcmaster.ca)

Version: 4 Date: 1 Jun 2004

Dear editor,

The only changes made to the manuscript were the additions of the postal codes and email addresses for both authors and the indication of the corresponding author all made to the title page of the manuscript.

Yours truly,
Nancy Wilczynski